Author’s response to reviews

Title: Polycystic ovary syndrome is a risk factor for sarcopenic obesity: A case control study

Authors:
Laura McBreairty (l.e.mcbreairty@gmail.com)
Philip Chilibeck (phil.chilibeck@usask.ca)
Julianne Gordon (julianne.gordon@usask.ca)
Donna Chizen (donna.chizen@usask.ca)
Gordon Zello (gordon.zello@usask.ca)

Version: 4 Date: 13 May 2019

Author’s response to reviews:

Responses to reviews:

We thank the editor for providing further review of our manuscript. We have edited the manuscript as requested and provided responses to each comment below.

Editor Comments:

1. Thank you for stating in the Methods section that patients provided written informed consent. Please also include this in the Ethics approval section of the Declarations.

Response: This is included in the Ethics approval section of the Declarations (lines 288-289)

2. We noticed that the email addresses listed in the manuscript for some of the authors did not match those listed in the submission system. Please ensure that the email addresses for all authors are the same in both.

Response: We have corrected this

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colors. All
relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have removed the highlights of text that was revised